Author’s response to reviews

Title: Does Presence of Metabolic Syndrome Impact Anxiety and Depressive Disorder Screening Results in Middle Aged and Elderly Individuals? A Population Based Study

Authors:

Jurate Butnoriene (jureatebutnoriene@yahoo.com)
Vesta Steibliene (vsteibliene@gmail.com)
Ausra Saudargiene (ausra.saudargiene@vdu.lt)
Adomas Bunevicius (a.bunevicius@yahoo.com)

Version: 2 Date: 04 Dec 2017

Author’s response to reviews:

December 4, 2017

Dear Editor,

Please find attached the manuscript "Does Presence of Metabolic Syndrome Impact Anxiety and Depressive Disorder Screening Results in Middle Aged and Elderly Individuals? A Population Based Study" after essential language revision, suggested by the reviewers - for publication in BMC Psychiatry.

This revision of the manuscript has been approved by all other co-authors of the manuscript.

We look forward to hearing from you.

Yours sincerely,

Jurate Butnoriene, MD, PhD